Author's response to reviews

Title: Purchasing and Using Personal Emergency Response Systems (PERS): How Decisions are Made by Community-Dwelling Seniors in Canada

Authors:

Alexandra C McKenna (mckenna.alexandra@gmail.com)
Marita Kloseck (mkloseck@uwo.ca)
Richard G Crilly (richard.crilly@sjhc.london.on.ca)
Jan Polgar (jpolgar@uwo.ca)

Version: 6 Date: 4 June 2015

Author's response to reviews:

Dear Dr. Pugh,

Thank you for accepting this manuscript for publication.

We have made final revisions as requested. Please find our point-by-point responses below:

1. This project received no funding - thus nothing needs to be added to the acknowledgements section.
2. I have included an abbreviations section as requested. This follows the abstract in the manuscript.
3. We have removed the figure, and all references to the Figure 1, from the manuscript as requested.

With thanks,

Marita Kloseck, PhD